Author's response to reviews

Title: Cytotoxicity Screening of Bangladeshi Medicinal Plant Extracts on Pancreatic Cancer Cells

Authors:

Sherine George (sGeorge4@illinois.edu)
Siddharth V Bhalerao (sbhaler2@illinois.edu)
Erich A Lidstone (elidst2@illinois.edu)
Irfan S Ahmad (isahmad@illinois.edu)
Atiya Abbasi (atiya.abbasi@iccs.ed)
Brian T Cunningham (bcunning@illinois.edu)
Kenneth L Watkin (watkin@illinois.edu)

Version: 6 Date: 13 September 2010

Author's response to reviews: see over
September 12th, 2010
Biomed Central Editorial Board

Dear Editors,

**SUB: Manuscript revisions**

Please accept our revised manuscript (no. 1348472814366240) titled “Cytotoxicity Screening of Bangladeshi Medicinal Plant Extracts on Pancreatic Cancer Cells”.

We have corrected two typographical errors (indicated in red) in the manuscript. We have also changed the x-axis range from 0.1-1000 ug/mL to 0.01-1000 ug/mL in figure 4B to maintain consistency across the three subfigures.

Thanking you,
Sincerely
Kenneth L. Watkin
(Corresponding Author)